Author's response to reviews

Title: The Alberta population-based prospective evaluation of the quality of life outcomes and economic impact of bariatric surgery (APPLES) study: background, design and rationale

Authors:

Raj S Padwal (rpadwal@ualberta.ca)
Sumit R Majumdar (me2.majumdar@ualberta.ca)
Scott W Klarenbach (swk@ualberta.ca)
Dan W Birch (dbirch@ualberta.ca)
Shahzeer Karmali (Shazeer.Karmali@albertahealthservices.ca)
Linda McCargar (Linda.McCargar@ales.ualberta.ca)
Konrad Fassbender (konrad.fassbender@ualberta.ca)
Arya M Sharma (amsharm@ualberta.ca)

Version: 5 Date: 5 October 2010

Author's response to reviews:

Reference 42: please provide a functioning web link to the website. DONE

References: please provide the full citation details for reference xxx (In Press). THIS WAS REFERENCE 30 AND IS FIXED

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. TRIED TO DO THIS _ LOOKS OK TO ME

References: the reference list should contain up to 30 authors' names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors. DONE

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance. DONE

Thank you